Journal of Hydrology 560 (2018) 259-274

Contents lists available at ScienceDirect

Journal of Hydrology

journal homepage: www.elsevier.com/locate/jhydrol





# How have the river discharges and sediment loads changed in the Changjiang River basin downstream of the Three Gorges Dam?



HYDROLOGY

Leicheng Guo<sup>a,\*</sup>, Ni Su<sup>b</sup>, Chunyan Zhu<sup>a,c</sup>, Qing He<sup>a</sup>

<sup>a</sup> State Key Lab of Estuarine and Coastal Research, East China Normal University, North Zhongshan Road 3663, Shanghai 200062, China
<sup>b</sup> State Key Lab of Marine Geology, Tongji University, Siping Road 1239, Shanghai 200090, China
<sup>c</sup> Faculty of Civil Engineering and Geosciences, Delft University of Technology, Delft 2628 CN, The Netherlands

#### ARTICLE INFO

Article history: Received 20 August 2017 Received in revised form 8 March 2018 Accepted 14 March 2018 Available online 15 March 2018 This manuscript was handled by C. Corradini, Editor-in-Chief, with the assistance of Georgia Destouni, Associate Editor

Keywords: River discharge Streamflow Sediment loads Changjiang Three Gorges Dam

# ABSTRACT

Streamflow and sediment loads undergo remarkable changes in worldwide rivers in response to climatic changes and human interferences. Understanding their variability and the causes is of vital importance regarding river management. With respect to the Changjiang River (CJR), one of the largest river systems on earth, we provide a comprehensive overview of its hydrological regime changes by analyzing long time series of river discharges and sediment loads data at multiple gauge stations in the basin down-stream of Three Gorges Dam (TGD). We find profound river discharge reduction during flood peaks and in the wet-to-dry transition period, and slightly increased discharges in the dry season. Sediment loads have reduced progressively since 1980s owing to sediment yield reduction and dams in the upper basin, with notably accelerated reduction since the start of TGD operation in 2003. Channel degradation occurs in downstream river, leading to considerable river stage drop. Lowered river stages have caused a 'draining effect' on lakes by fostering lake outflows following TGD impoundments. The altered river–lake interplay hastens low water occurrence inside the lakes which can worsen the drought given shrinking lake sizes in long-term. Moreover, lake sedimentation has decreased since 2002 with less sediment trapped in and more sediment flushed out of the lakes. These hydrological changes have broad impacts on river flood and drought occurrences, water security, fluvial ecosystem, and delta safety.

© 2018 Elsevier B.V. All rights reserved.

# 1. Introduction

River and sediment discharges in worldwide large rivers are subject to significant changes (dominantly reduction) because of climate changes and increasing human activities (Walling and Fang, 2003; Syvitski et al., 2005). Global climate changes induce accelerated hydrosphere circulation, resulting in increased frequency of extreme events (e.g., droughts and floods) and possibly evoking hydrological and ecological influences at a broader scale (Xu and Singh, 2004; IPCC, 2013; WMO, 2016). Arnell and Gosling (2013) had modeled and projected that climate changes will cause significant changes in hydrological behavior by 2050, e.g., streamflow increase for nearly half of the global land surface and decrease over the other 36% of the land surface. Regional variability is still high because of the uncertainties in the climate models and the non-linear regional responses to global changes (Arnell and Gosling, 2013; Nakaegawa et al., 2013). Regional case studies

\* Corresponding author. *E-mail address:* lcguo@sklec.ecnu.edu.cn (L. Guo). can provide perspectives for interpretation of hydrological and climatic changes at global scale.

Human activities in terms of dam constructions, water consumption, and land use changes exert increasing influences on basin-scale hydrological regimes (Nilsson et al., 2005; Syvitski et al., 2005). Dams are built worldwide to control river floods and store water for irrigation and power generation. On the other hand, dams can significantly alter hydrological, geomorphological, and ecological processes in fluvial systems (Milliman, 1997; Brandt, 2000; Walling and Fang, 2003; Graf, 2006). For instance, at basin scales, dam operation regulates river discharge hydrograph by changing the timing, magnitude and frequency of low and high flows, and disrupts sediment delivery, e.g., the Aswan Dam in the Nile River (Woodward et al., 2007) and the Hoover Dam in the Colorado River (Graf, 2006). At continent scales, large artificial reservoirs are able to alter regional climate regarding changes in precipitation and surface evaporation etc., particularly in the semi-arid regions (Degu et al., 2011). Moreover, increasing global water storage by man-made reservoirs has reduced water volume reaching oceans, potentially slowing down global sealevel rise by 30 mm in the past half century (Chao et al., 2008).



River flow regulation and irrigation at global scales also increase evapotranspiration and water consumption (Jaramillo and Destouni, 2015). These evidences demonstrate remarkable regional to global effects of water usage and hydrological regime changes, which merit specific examinations.

The Changjiang River (CJR, also called Yangtze River) basin in China is such a system undergoing significant hydrological changes under the influences of both climatic changes and human interferences. Regional climatic changes include changing intensity and behavior of the Asian monsoon (Xu et al., 2011a), while human activities include increasing water consumption by a booming population, more hydropower dams and water transfers etc. A number of reports have been devoted to examining changes of precipitation, temperature, river and sediment discharges in the CIR basin. Yang et al. (2004, 2006), Gao et al. (2013), Wei et al. (2013), and Chen et al. (2001b, 2016) etc. had examined river discharge changes and reported no significant decrease or increase changes at the half century time scale. Yang et al. (2004, 2006, 2011, 2018), Wang et al. (2007b, 2008), Hu et al. (2009), and Xu et al. (2009) etc. had documented radical sediment load reductions due to the TGD. Li et al. (2013a, 2016), Dai and Lu (2013), Xu et al. (2013a), and Zheng (2016) discussed comprehensive impacts of the TGD on the river and sediment discharges and associated ecological, environmental and socio-economic influences. These studies provide useful insights on the hydrological regime changes. On the other hand, we find that there is inconsistency remained to the author's knowledge. For instance, Yang et al. (2006), Hu et al. (2009) and Zhao et al. (2017) documented no apparent decrease or increase trend of the annual streamflow at Yichang and Datong (data range 1953–2013), whereas Jiang et al. (2007) detected a weak decrease trend at Yichang and a weak increase trend at Datong (data range 1961-2005). The different interpretations maybe caused by inconsistent data range and/or different analysis methods at different significance levels. In addition, most previous studies interpreted inter-annual changes of streamflow and sediment loads while the seasonal changes were insufficiently examined. Therefore, there is still a demand to clarify hydrological changes by rigorous trend analysis of datasets at annual and seasonal time scales.

Additionally, low water conditions (low river discharges and low river stages) frequently occurred in the middle-lower CJR basin downstream of TGD in the recent decade. For instance, nearly the lowest or the second lowest river stages at a 50-150 year time scale were recorded downstream of Cuntan (Fig. 1) in the summer of 2006 (Dai et al., 2011). Regional low water occurred successively in 2006, 2009, 2011, and 2013 in the Dongting and Poyang lakes (Fig. 1; Li et al., 2012). This is controversial to that the TGD discharges more water in the dry seasons which is expected to relieve low water occurrence. Big floods still occurred in the downstream basin after the TGD was constructed, rising doubt on the effectiveness of the TGD in mitigating flooding risk. There are increasing debates on TGD's negative influences on the downstream basin regarding water resources security, and drought and flood hazard, and fluvial ecosystems. The TGD has been put into operation for 14 years since 2003 and a rigorous assessment of its impacts on the river discharge and sediment delivery processes is needed for integrated river management.

In this work we review a wide range of publications, collect a long time series of river discharges and sediment loads data at multiple gauge stations in the basin downstream of the TGD, and examine them rigorously by trend analysis methods. We will clarify the TGD's role in the hydrological changes and elaborate consequent impacts in the river system. In the end we identify knowledge gaps in our present understandings. The rest of this paper is organized as follows. Section 2 briefly introduces the CJR basin, data source, and methods. Section 3 presents the operation of the TGD. Section 4 addresses changes with respect to hydrometeorology, river and sediment discharges at both seasonal and annual time scales, and Section 5 discusses causes and implications of the changes with a focus on the river–lake interactions. The last section formulates conclusions and briefs knowledge gaps.

# 2. Setting, data, and method

# 2.1. The Changjiang River basin

The CJR, stretching west-eastward in the middle China, is one of the world longest rivers of great ecological and socio-economic importance. It has a mainstem length of ~6,300,000 m, a basin area of ~1900 billion m<sup>2</sup> (19.5% of China's land area), an annual streamflow of 903.4 billion m<sup>3</sup> (37% of China's total streamflow) and an annual sediment load of 414 million tons (between 1950 and 2005 at Datong) (Yang et al., 2006; Wang et al., 2007b). The CJR is home to ~480 million people, indicating its critical role on socio-economic development (Chen et al., 2001b; Xu et al., 2008).

The CJR basin is geographically divided into the upper, middle, and lower sub-basins, and the delta regions with divisions at Yichang, Jiujiang, and Datong, respectively, based on landscape (Fig. 1). The upper basin has a high relief and is mountainous and the main sediment source region, while the middle-lower basin consists of vast fluvial flood plains and its mainstream channel is a conduit conveying sediment to the sea. The TGD is about 40 km upstream of Yichang. Upstream of the TGD, Cuntan is the main gauge station where the TGD inflows are monitored. Another new dam, the Xiangjiaba Dam, which has a flood control capacity of 0.9 billion  $m^3$  and is  ${\sim}408 \mbox{ km}$  upstream of Cuntan, was constructed and put into operation since 2012. Downstream of the TGD, Yichang, Shashi, Jianli, Luoshan, Hankou, Jiujiang, and Datong are the major gauge stations along the mainstream in the downward direction (Fig. 1). The annually mean river discharges are 10,870, 13,620, 28,400 m<sup>3</sup>/s at Cuntan, Yichang and Datong, respectively.

A number of big tributaries scatter in the CJR basin, such as the Wu River, the Han River, and the Dongting Lake and Poyang Lake systems (Fig. 1). The Dongting Lake receives flows from four secondary tributaries and conjuncts the CJR mainstream at Chenglingji. In addition, there are three inlets diverting mainstream water and sediment into the Dongting Lake in the wet seasons (Fig. 1). The Poyang Lake has five secondary tributaries and conjuncts the CJR mainstream at Hukou. Reverse flows, i.e., flows from the CJR mainstream to the Poyang Lake, can happen at Hukou occasionally in the wet seasons. The water and sediment exchanges between the lakes and mainstream play a profound role in buffering hydrological changes.

The CJR basin has a basin-wide mean annual precipitation of ~1100 mm and the precipitation exhibits strong spatiotemporal variations owing to Asian monsoon (Wang et al., 2012). The mean annual precipitation is 270-500 mm in the upper basin and 1,600-1900 mm in the middle-lower basin (Gemmer et al., 2008). Temporally, rainfall concentrates in the wet seasons between May and September and as a result a major portion of streamflow (~75%) and sediment loads (~85%) are discharged in the wet seasons (Chen et al., 2001b). Spatially, the prime rainy season is between May and October in the upper CJR basin, one-to-two month lagging the rainy season in the middle-lower basin, i.e., between March and August (Jiang et al., 2007; Guo et al., 2011). Extreme floods or droughts are likely to happen if wet or dry climate occurs occasionally in the same season between the upper and middle-lower basins, such as the big river floods in the summer of 1954 and 1998 (Zong and Chen, 2000) and the severe droughts in the



**Fig. 1.** Sketch maps of (A) the location of the CJR basin in China, (B) the CJR basin as a whole, (C) the middle and lower CJR basin with major tributaries and the gauge stations, and (D) a cartoon showing the relative position of the tributaries and gauge stations. The TGD and TGR mean the Three Gorges Dam and the Three Gorges Reservoir, respectively. The CJRD refers to the Changjiang River delta.

autumn of 2006 (Dai et al., 2011). The highest flood peak discharges ever recorded are 105,000 m<sup>3</sup>/s at Yichang (in 1870) and 92,600 m<sup>3</sup>/s at Datong (in 1954). Catastrophic floods impose huge risks on the densely populated flood plains downstream of Yichang and there was an idea of constructing the TGD to control big floods many decades ago.

# 2.2. Data

We collect hydro-meteorological data including air temperature, precipitation, daily river discharges and water levels, monthly and yearly streamflow and sediment loads at multiple stations along the mainstream downward Cuntan and at the mouths of the tributaries and lakes (Fig. 1). Air temperature and precipitation data are collected from publications (NCC, 2011; Sang et al., 2012; Wang et al., 2012; Zhang et al., 2014). Daily river discharges and water levels (http://xxfb.hydroinfo.gov.cn) and the inflow and outflow of the TGD (http://www.ctgpc.com.cn) are obtained from official websites. The monthly and yearly streamflow and sediment loads are from government bulletins, e.g., CWRM (2006, 2016). The annual streamflow and sediment loads data date back to 1950s and the daily river discharges and water levels date back to 1980s. Annually mean river discharges, annually minimum and maximum monthly river discharges are also collected at Yichang (1878-1986), Hankou (1865-1986), and Datong (1948-1986), from a global river discharge data base (http://www.compositerunoff.sr.unh.edu/) (Fekete et al., 2002). Preliminary analysis suggest that the data from the global data base are consistent with the officially published data in the overlap period (1950-1986), with a mean deviation <1%. Note that the annually minimum monthly discharge is not the same as the minimum 30-day averaged discharge. The river discharge and sediment loads differences between the inflow (that at Cuntan plus Wulong) and outflow (that at Yichang) of the TGD are used to indicate its regulation effects. The streamflow differences between the inflow and outflow during TGD's impounding periods are taken as the amount of water stored by the reservoir when assuming limited water loss to evaporation and groundwater etc. Only surface flow is taken into account while groundwater and its exchange with surface water and evaporation are not analyzed in this work as a first approximate.

Sediment loads are represented by suspended sediment transport while bed load is excluded in this study. It is because bed load transport accounts for a small proportion (<5%) of the total load transport. At Cuntan, for instance, the annual bed load is 0.1–0.6 million tons prior 2000 and is <0.1 million tons afterward (Fang and Dong, 2011). The bed load transport flux is much smaller compared to suspended load downstream of Yichang (Chen et al., 2010). It is technically difficult to monitor bed load transport accurately in fluvial sandy environments and data in that sense is scarce.

# 2.3. Trend analysis methods

We first apply change-point analysis (Taylor, 2000) to the normalized discharge anomaly to test non-linear trend (e.g., monotonic non-linear, plateau, through, peak, rapid changes). The cumulative function S, whose values S<sub>i</sub> are defined as:  $S_i = S_{i-1} + (\overline{X} - X_i)$  while  $X_i$ , i = 1, 2...N, is the time series of normalized anomaly and  $\overline{X}$  is the mean value. A potential change point is detected when a change in the direction of the S function occurs, whose interval of occurrence is defined by the parameter  $S_{diff} = max(S) - min(S)$ . Statistical significance of the identified change points is verified by using a bootstrapping technique at a 5% significance level (Castino et al., 2016). It is achieved by randomly resampling (to remove any potential change point) M times  $(M > 10^3)$ , in this analysis  $10^4$ ) and the statistics for the parameter S<sub>diff</sub> is generated. The level of significance is given by the percentage of bootstrap cycles for which the synthetic S<sub>diff</sub> parameter is greater than the observed one. Once a primary, statistically significant change point is obtained, the same procedure is applied to the sub-time series of the X variable before and after the change point, to search for secondary change points. This iterative procedure is applied as long as statistically significant change points are detected (Castino et al., 2016). Preliminary tests show that the change-point timing is independent of the normalization reference period thus the mean value in the interval of 1950-1985 (considering that the mid-1980s is a visual changing point of annual streamflow and sediment load at most stations, see Section 4 for details) is used in this study if not specified in particular. Note that the detected change points may not locate at an exact year but around some year when the changes are not sharp.

We then apply linear regression analysis to the normalized anomaly of streamflow and sediment loads to detect linear trends and change rates on both sides of the change points. Student's *t* test and Mann-Kendall (MK) test at 5% significance level are used to determine the statistical significance of the linear trend (Mann, 1945; Kendall, 1975). Details of the linear trend analysis methods can be found in literatures thus are not repeated here.

# 3. TGD operation

In this section we provide a brief overview of TGD operation and its effects on river discharge hydrograph. The TGD was constructed from 1993 to 2009. With a dam height of 185 m (above mean sea level) and a dam length of 2335 m, the TGD forms a reservoir stretching about 600,000 m between the dam site and Cuntan (Fig. 1) and with a surface area of 1084 million m<sup>2</sup> (which was 452 million m<sup>2</sup> under no TGD situation), a total storage capacity of 39.3 billion m<sup>3</sup>, and a flood control capacity of 22.15 billion m<sup>3</sup> at a high pool level of 175 m. The flood control capacity is ~5% of the annual streamflow at Yichang, i.e., annually 451.0 billion m<sup>3</sup> between 1950 and 2005. The TGD has generated a hydropower output of averagely 88.2 billion kilowatt hours per year in the past 14 years. Flood control, power generation, and promoting irrigation and navigation are its essential functions.

The TGD operation follows a scheme by storing and flushing water seasonally. Rising water level ahead the dam (WLAD) suggests larger inflow than outflow and TGD impoundments while falling WLAD indicates larger outflow than inflow and water release (Figs. 2 and S1). Specifically, the WLAD maintains at a low level of 145 m between June and September (the wet season). sustaining the flood control volume available for occasional river floods forming in the upper basin with a peak discharge >55.000  $m^3/s$  (Zheng, 2016). Since early October when the wet season ends. the TGD will reduce outflow and store water for hydropower generation until the WLAD reaches 175 m. The WLAD will stay at 175 m until January after which the WLAD drops gradually until June when the wet season begins. Overall the TGD operation reflects a flood-control scheme which is designed to control river floods and to store water for hydropower generation (Zheng, 2016). Regarding reservoir sedimentation management, the TGD operation reflects a strategy of 'storing clear water (in the dry season with a high pool level) and releasing muddy water (in the wet seasons with a low pool level)'. Even though reservoir sedimentation is still huge in its early stage of lifespan because of a large reservoir size (see Section 5.3).

The realistic TGD operations follow the abovementioned scheme other than that the starting time of its impoundments is 1–2 months earlier than October (Fig. 2). Specifically, the TGD started to store water the first time in May of 2003 and the WLAD was raised from 67 to 135 m. In the second phase of its pilot operation, the WLAD was raised from 135 to 156 m in October of 2006. In the third phase, the WLAD was raised from 145 to 172 m in 2008. Given concerns of the safety of the dam and the influences on the downstream basin, the WLAD did not reach 175 m until 2010 the first time (Fig. 2). Since then the TGD is put into normal operation with a seasonally and regularly raised and dropped pool



Fig. 2. Yearly variations of water level ahead the dam (WLAD) in comparison with the designed operation scheme.

level. So far we see that TGD impoundments in autumn last for a period of 37–90 days (Table 1).

We compare TGD's inflow and outflow hydrographs to quantify its impacts on river discharges. It shows that the peak river discharges of the outflow are up to 20,000 m<sup>3</sup>/s smaller in a short period of days to weeks, reflecting TGD's capability in mitigating big floods (see Fig. S2 in the Appendix). The outflow during the impounding periods in autumn is averagely reduced by 2790– 5880 m<sup>3</sup>/s (Table 1). By adding the stored water to the actually measured streamflow at Yichang and Datong during the impounding periods, we estimate that the mean streamflow is reduced by 11–32% and 7–20%, respectively. By contrast, the outflow is averagely 2000 m<sup>3</sup>/s larger than the inflow in the dry months between January and May (see Fig. S2). Overall the TGD modulates river discharge hydrograph by reducing flood peak discharges and river discharges during the impounding periods, and slightly increasing river discharges in the dry seasons.

# 4. Results

# 4.1. Hydro-meteorological changes

This section describes hydro-meteorological changes in terms of air temperature, precipitation, and evapotranspiration. The nation-wide daily maximum temperature is increasing in China since 1980s owing to changes in the Asian monsoon, indicating a warm trend (Kuang et al., 2014). In the CJR basin, the basin-wide mean air temperature decreases slightly between 1950 and 1985, followed by a fast increase trend indicating climatic warming as well (Figs. 3A and S3A; Chen et al., 2001a; Sang et al., 2012). Linear trend analysis reveals an increase rate of 0.45 °C per decade between 1985 and 2010 for the CJR basin as a whole (Figs. 3A and S3A).

The precipitation exhibits large spatial and temporal variations in the CJR basin (Wang et al., 2012; Zhang et al., 2014; Chen et al., 2016). The basin-averaged precipitation shows a significant decrease trend between 1950 and 2015 (p < 0.02), i.e., a decrease rate of ~13 mm per decade (Figs. 3B and S3B). Similar decrease at decadal time scales occurs in the upper CJR basin particularly in the interval of 1950–1970 while the afterward decrease trend is non-significant. Seasonally, summer precipitation displays an increase trend while autumn precipitation decreases significantly particularly since the mid-1980s (Becker et al., 2006). In contrast, a significant decrease of summer precipitation occurs in the upper CJR basin (Jiang et al., 2007; Gemmer et al., 2008; Deng et al., 2013). The basin-wide precipitation intensity in short durations (Becker et al., 2006; Zhang et al., 2014).

Regarding evaporation, both annual reference evapotranspiration and pan evaporation show a significant decrease trend between 1960 and 2007, which is ascribed to decreasing net total radiation and wind speed (Xu et al., 2006; Wang et al., 2011). The decrease in evapotranspiration is more significant in summer and in the middle-lower basin (Wang et al., 2007a, 2011). In the middle-lower CJR basin, the annual reference evapotranspiration also decreases significantly between 1960 and early-1990s mainly due to declined net radiation, followed by an afterward slight increase which is ascribed to reduced relative humidity in the air (Xing et al., 2016).

As an evaporation source, the large reservoir formed by TGD potentially has impacts on regional climate considering its surface cooling effects and reduced radiative heating. The impacts diminish fast with distance away from the reservoir and so far measurable but non-significant air temperature decrease is detected in the regions surrounding the reservoir (Miller et al., 2005; Wu et al., 2012). Data and modeling results suggest that the reservoir's impact on basin-scale climate is overall limited (NCC, 2011; Wu et al., 2012). Further monitor and in-depth examination is needed to clarify the degree to which the large reservoirs formed by TGD and other new big dams in the upper CJR basin can affect regional and sub-basin climate.

When looking forward into the future, Sun et al. (2013) modeled that the temperature will increase and the annual rainfall will decrease in the upper CJR basin in the 21st century, as a result of climatic warming. Nakaegawa et al. (2013) predicted that the streamflow of the CJR basin may increase in the wet seasons between May and August if considering climate change impacts only. Considering the uncertainties in the model projections (Birkinshaw et al., 2017) and the land use changes in the future, it remains an open question how the basin-scale hydrometeorology will change in response to climate changes and human activities.

# 4.2. River discharge changes

#### 4.2.1. Annual streamflow changes

The annual streamflow exhibits only slight decrease trends along the mainstream downward TGD. At Cuntan, a nonsignificant change point (p < 0.11) is detected in 1968 based on data between 1940 and 2015 (Figs. 4A and S4A). The change points detected in 1954 and 1968, however, are significant (p < 0.02) at Yichang (Figs. 4B and S4A). There are no significant directional changes prior 1954 but an afterward decrease trend by a rate of 4.8 billion m<sup>3</sup> per decade between 1955 and 2015. Over century, the annual streamflow decreases from averagely 455.8 billion m<sup>3</sup> in 1878–1954 to 402.6 billion m<sup>3</sup> in 2003–2015 at Yichang, indicating a reduction by 11.7%. At Hankou, the annual streamflow decreases from averagely 750.2 billion m<sup>3</sup> in 1865–1954 to 696.7 billion m<sup>3</sup> in 1955–2015, but the change point detected in 1954 is non-significant (p < 0.21) (Figs. 4C and S4B). Similarly, the

Table 1					
TGD impoundments in	the recent year	ars and thei	r impacts on	reducing rive	r discharges.

Year	Impounding period	Stored water (km <sup>3</sup> )	Mean discharge reduction (m <sup>3</sup> /s)	Streamflow at Yichang (km <sup>3</sup> )	Streamflow at Datong (km <sup>3</sup> )
2003	24 May-10 Jun	10.0	6430	10.78	56.04
2006	21 Sep-27 Oct	11.1	3470	30.52	50.15
2008	28 Sep-04 Nov	19.3	5880	40.25	82.78
2009	15 Sep-30 Oct	17.7	4440	40.56	70.69
2010	22 Aug-26 Oct	21.0	3800	103.7	204.5
2011	20 Aug-30 Oct	21.8	3550	67.45	135.9
2012	21 Aug-30 Oct	21.7	3540	112.7	214.5
2013	27 Aug-13 Nov	20.9	3100	73.8	140.9
2014	02 Aug-31 Oct	21.7	2790	182.5	282.7
2015	17 Aug-01 Nov	21.4	3260	114.4	183.2
2016	08 Sep-02 Nov	21.5	4430	50.7	98.6



Fig. 3. Variations of the normalized anomaly of (A) annual temperature (NAT) and (B) annual precipitation (NAP) averaged in the upper CJR basin and in the entire basin as a whole.



**Fig. 4.** The normalized anomaly of the annual streamflow (NAQ) and associated cumulative functions (CSF) at (A) Cuntan (1940–2015), (B) Yichang (1878–2015), (C) Hankou (1865–2015), and (D) Datong (1947–2015). The dots indicate the detected change points.

change point in 1954 is non-significant at Datong though the data length is shorter (Figs. 4D and S4C). The change point in the early 2000s is non-significant at both Yichang and Datong (p < 0.18), suggesting limited impacts of the TGD on the annual streamflow. In the interval of 1955–2015, the decrease trends of the annual streamflow are significant at Cuntan (p < 0.01) and Yichang (p < 0.02), whereas it is non-significant at Hankou (p < 0.08, 0.6 billion m<sup>3</sup> per decade) and Datong (p < 0.14, 6.9 billion m<sup>3</sup> per decade).

More temporal streamflow changes are detected in the tributaries based on data between 1950 and 2015. In the Dongting Lake system, the total streamflow of the three inlets exhibits significant decrease change points in the late 1960s (p < 0.01) and early 2000s (p < 0.05) (Figs. 5A and S5A). Specifically, it decreases from 134.2 billion m<sup>3</sup> in 1956–1968 to 48.6 billion m<sup>3</sup> in 2001–2015, a reduction by 64%. With no directional changes of the total streamflow of



**Fig. 5.** The normalized anomaly of the annual streamflow (NAQ) (A) of the Dongting (DT) Lake tributaries, at three inlets, and at Chenglingji, (B) of the Poyang (PY) Lake tributaries and at Hukou, and (C) at Huangzhuang and the total of the three tributaries between 1950 and 2015.

the four tributaries, the streamflow at Chenglingji has decreased significantly between 1950 and 1970 (p < 0.003), followed by a non-significant decrease trend until 2015, i.e., from 338.3 billion m<sup>3</sup> in 1951–1968 to 241.3 billion m<sup>3</sup> in 2001–2015. No significant change points are detected on the time series of annual streamflow at Huangzhuang, Hukou, and Poyang Lake tributaries (Fig. 5B and C). The total streamflow of the Poyang Lake tributaries and that at Hukou exhibits consistent temporal variations (Figs. 5B and S5B). Linear trend analyses suggest significant streamflow decrease trends at Chenglingji (p < 0.0001), three inlets (p < 0.0001), and Huangzhuang (p < 0.03) in 1950–2015, while the decrease trends for Dongting Lake tributaries (p < 0.81) and the increasing trends at Hukou (p < 0.60) and Poyang Lake tributaries (p < 0.66) are non-significant. In all, the total streamflow of the Dongting Lake, the Han River, and Poyang Lake, represented by the summed streamflow at Chenglingji, Huangzhuang, and Hukou minus that at three inlets, exhibits a non-significant increase trend (p < 0.27) (Figs. 5C and S5C).

#### 4.2.2. Seasonal river discharge changes

River discharge may change remarkably at intra-annual time scales when the annual streamflow remains unchanged. The annually minimum monthly discharges (not necessarily in the same month in different years) increase substantially in the recent decades. The minimum monthly discharges are slightly smaller in 1940–2000 compared to 1880–1940 at Yichang (Figs. 6A and S6A). A significant change point is present at both Cuntan and Yichang in 1999. The abrupt increase at Yichang since 2007 is ascribed to TGD operation while the abrupt increase at Cuntan since 2012 is caused by Xiangjiaba Dam. Specifically, the annually minimum monthly discharges have increased by up to ~3000 m<sup>3</sup>/s at Yichang, while the differences between Cuntan and Yichang increase sharply from averagely ~800 m<sup>3</sup>/s (1940–1999) to ~1900 m<sup>3</sup>/s (2007–2015). It suggests that the TGD flushes nearly ~2000 m<sup>3</sup>/s more water downstream in the driest months.

The annually minimum monthly discharge starts to increase since 1900s at Hankou, and the afterward increase rate becomes larger with a significant change point in 1954 (p < 0.001) (Fig. S6B). Similar increase is observed at Datong since the late

1950s (Figs. 6A and S6C), i.e., an increase from averagely 9260  $m^3/s$  (1960–1980) to 12,000  $m^3/s$  (2007–2015). The changes since 2007 are less noticeable at both Hankou and Datong compared to Yichang, suggesting downward dissipated changes.

The annually maximum monthly discharge exhibits more spatial and temporal changes. It changes limitedly at Cuntan since 1960 whereas a stepwise reduction in late 1950s and a linear reduction trend since 1980s are detected at Yichang (Figs. 6B and S7A). The maximum monthly discharges are averagely  $\sim$ 3400 m<sup>3</sup>/s smaller at Yichang in 2001–2015 compared to 1955–2000. They are slightly smaller in the decades since 2000 compared to the prior decades at Hankou and Datong (Figs. S7B and S7C). Higher streamflow in the 1990s affects the trend detection at Hankou and Datong, and it is too early to conclude whether the decrease trend since 2000 is significant.

The mean river discharges in October exhibit decrease trends. The October river discharges decrease slightly since 1980 at Cuntan (Figs. 6C and S8A). The decrease trend is at a faster rate at Yichang than Cuntan in the decades since 1980 and the October river discharges become smaller at Yichang since 2006, implying



Fig. 6. The normalized anomaly (NAQ) of (A) annually minimum monthly, (B) annually maximum monthly discharge, (C) monthly river discharge in October at Cuntan, Yichang, and Datong in the periods of 1960–2015. The normalization reference period is chosen as the period prior 1960.

the impacts of TGD impoundments in autumn. Specifically it declines from 19,800 m<sup>3</sup>/s in 1890–1954 to 11,000 m<sup>3</sup>/s in 2006–2015 at Yichang. Similarly, the October river discharges reduce significantly since 1960s at both Hankou and Datong (Figs. S8B and S8C), e.g., from 33,600 m<sup>3</sup>/s in 1960–1980 to 23,800 m<sup>3</sup>/s in 2006–2015 at Datong.

We further estimate the probability distributions of daily river discharges at Cuntan, Yichang, and Datong to infer seasonal changes. Notably we find that the probability of both extremely low and high river discharges has decreased while that of medium river discharges (e.g., river discharges in the range of 10,000-25,000 m<sup>3</sup>/s at Datong) has increased at all three stations (Fig. 7A). The river discharge at the 50th percentile reduces accordingly (Fig. 7B). The mean river discharge hydrographs show that the low river discharges in the dry seasons slightly increase while the high river discharges in the wet seasons decrease profoundly (see Fig. S9). The river discharge reduction is much more significant at Yichang and Datong than Cuntan between August and October, again demonstrating the impacts of TGD impoundments. Overall, these results suggest that the river discharge hydrograph of TGD inflow is ever changing and the TGD operation modulates it at a seasonal time scale.

# 4.3. Changes of sediment loads

### 4.3.1. Sediment loads changes along the mainstream

The sediment loads in the CJR basin exhibit radical decreases at the decadal time scales, particularly since 1980s. Significant change points are detected in the early 1990s (p < 0.004) and early 2000s (p < 0.012) at both Cuntan (plus Wulong to represent sediment influx of TGD) and Yichang (Figs. 8A and S10). The sediment loads variations display consistent behavior between Cuntan and



**Fig. 7.** (A) the frequency distribution, and (B) the cumulative frequency distribution of the daily river discharges at Cuntan, Yichang, and Datong in two periods.



**Fig. 8.** The normalized anomaly of sediment loads (NAG) (A) at Cuntan (plus Wulong), Yichang, and Datong, (B) at three inlets, Chenglingji, and that at Chenglingji minus at three inlets, and (C) at Huangzhuang and Hukou. The shade bars indicate change points.

Yichang in the decades prior 2002, though it is slightly larger at Yichang. Since 2003, the sediment loads become smaller at Yichang and the differences between Cuntan and Yichang become larger, reflecting sediment trapping effects of the TGD. Specifically, the summed sediment loads at Cuntan and Wulong have not changed very much between 1950 and mid-1980s (annually  $487.6 \pm 231.9$ million tons), but decreased to  $164.4 \pm 131.3$  million tons in 2003– 2015, indicating a reduction by 66%. Immediately downstream of the TGD, the sediment loads are  $527.2 \pm 221.4$  and  $40.4 \pm 69.6$  million tons in 1950–1985 and 2003–2015, respectively, at Yichang, indicating a reduction by 92%. The dramatic reduction reflects combined effects of a source reduction upstream of Cuntan and TGD sedimentation. In all these changes suggest that the sediment loads have reduced progressively since the mid-1980s and the TGD accelerates the reduction since 2003 at a remarkable rate.

The sediment loads downstream of Yichang follow the reduction trend but at a smaller rate. At Hankou (not shown), the sediment loads have decreased from  $431.3 \pm 167.8$  (1953-1985) to  $105.9 \pm 68.1$  million tons (2003–2015), suggesting a reduction by 75%. Further downstream, at Datong, it decreases from annually 470.4 ± 207.6 million tons in 1951–1985 to 138.7 ± 77.3 million tons in 2003-2015, suggesting a reduction by 70%. Over time, four statistically significant change points are detected in the late 1960s (p < 0.04), mid-1980s (p < 0.05), early 1990s (p < 0.02), and early 2000s (p < 0.001) at Datong. More variability at Datong than Yichang is ascribed to superimposed changes occurred in the upper and middle-lower CJR basins (see Section 5.3). In the near future decades, the annual sediment loads at Yichang are highly likely to remain at a low level (<10 million tons) considering ongoing dam constructions in the upper basin (Fig. S10). With sediment compensation from tributaries and bed erosion (see Section 5.3), the sediment loads at Hankou and Datong seem to have reached a decadal stable state, i.e., annually averaged  $\sim$ 88.2 and  $\sim$ 126.1

million tons, respectively, in 2006–2016. It is reasonably expected that the sediment loads will not restore to a mean value larger than that in future at Hankou and Datong, and further reduction may occur considering diminished river erosion downstream of TGD.

#### 4.3.2. Sediment loads changes in the main tributaries

The sediment loads of the main tributaries in the middle CIR basin have reduced profoundly as well. The total sediment loads of the four tributaries of the Dongting Lake system exhibit nonsignificant changes between 1950 and the mid-1980s (except the extreme peak in 1954) but decrease abruptly in the mid-1980s, followed by a slow decrease trend until nowadays (Figs. 8B and S11A). The annual sediment loads are  $35.2 \pm 23.5$ ,  $17.8 \pm 11.4$ , and  $8.0 \pm 4.5$  million tons in the interval of 1953–1984. 1985– 2005, and 2006-2015, respectively. At the three inlets, the sediment loads display three significant change points in the late 1960s, mid-1980s, and early 2000s. It decreases from averagely 200.2 ± 59.4 million tons in 1956–1968 to 118.2 ± 59.5, 64.7 ± 87.6, and 6.5 ± 6.8 million tons in 1969-1984, 1985-2005, and 2006-2015, respectively. Accordingly, significant change points of sediment loads reduction are detected in the early 1970s (p < 0.001), early 1980 (p < 0.008), and mid-2000 s (p < 0.001) at Chenglingji (Figs. 8B and S11A). The sediment loads have decreased from averagely  $62.5 \pm 22.5$  million tons in 1951–1970 to  $22.1 \pm$ 7.5 million tons in 2008–2015. Note that slightly more sediment is flushed out of the Dongting Lake at Chenglingji in 2008-2015 compared to the previous decade (Fig. S11A).

The sediment balance between the mainstream and the Dongting Lake is reversed around 2006. The CJR mainstream diverted more sediment to the lake via the three inlets than its gain from the lake via Chenglingji prior 2006 (Fig. S11A). The imbalance diminished over time due to more dramatic sediment loads reduction at the three inlets. Specifically, the CJR mainstream annually lost a net of 136.3  $\pm$  56.1 and 48.1  $\pm$  72.1 million tons of sediment to the Dongting Lake in 1956–1966 and 1990–2000, respectively. In the recent decade since 2006, the CJR mainstream has an annual sediment surplus of 16.1  $\pm$  7.7 million tons regarding its exchange with the lake (Fig. S11B). With respect to the Dongting Lake, lake sedimentation and sediment surplus occurs prior 2006 because of more sediment sources (from lake tributaries and three inlets) than loss (at Chenglingji) but it shifts to afterward lake erosion because of more sediment loss than the total sediment supply.

Similar changes are observed in the Poyang Lake system. The total sediment loads from five tributaries of the Poyang Lake show no directional changes between the mid-1950s and mid-1980s but a significant decrease trend (p < 0.001) between the late 1970s and late 2000s (Fig. S11C). It is averagely  $16.2 \pm 12.4$  million tons in 1956–1984 and  $6.0 \pm 6.9$  millions in 2001–2015. The sediment loads at Hukou have decreased gradually between 1970 and 2000 and displayed an abrupt increase in 2001 (p < 0.007) (Figs. 8C and S11C). The mean sediment loads are  $14.1 \pm 8.3$ ,  $6.2 \pm 5.6$ , and  $12.2 \pm 6.6$  million tons in the decades of 1952–1962, 1990–2000, and 2001–2015, respectively. As a result, the Poyang Lake receives less sediment supply from its tributaries comparing to its loss to the mainstream at Hukou since 2000, suggesting a shift from lake sedimentation to erosion as well.

The sediment loads of the Han River show two significant change points, i.e., in the late 1960s (p < 0.04) and late 1980s (p < 0.02) (Figs. 8C and S11D). The mean sediment loads are 119.4 ± 112.6 million tons in 1951–1964 but reduce sharply after the closure of the Danjiangkou Dam in the upper Han River in 1965, and decline further to a relatively small quantity in the recent decade, i.e., averagely 5.8 ± 11.3 million tons in 2001–2015.

The total sediment loads of the three main tributaries undergo dramatic decreases and exhibit strong temporal changes. From a sediment budget point of view, the CJR mainstream loses more sediment to the Dongting Lake than its total gain from the two lakes and the Han River prior 2001 (other than the extreme years) (Fig. S11D). The late 1960s is featured by maximal net loss due to dramatically reduced sediment loads from the Han River. The CJR mainstream attains a net sediment surplus in the recent decade since 2002, which is mainly because of reduced sediment loss via the three inlets and increased sediment loads at both lake mouths, i.e., Chenglingji and Hukou. The net sediment surplus is ~30.5 million tons per year in the interval of 2003–2015.

#### 4.3.3. Seasonal changes of sediment loads

Sediment loads also change profoundly at seasonal time scales. Basically >85% of annual sediment loads are discharged between May and October in the middle-lower CJR river, because of higher precipitation and associated higher river discharge and higher sediment yield in the wet seasons (Fig. 9). The sediment loads in the wet seasons, however, have decreased enormously in the recent decade since 2003 while the reduction in the dry seasons is less measurable. Specifically, the sediment loads in the wet season are 456.5  $\pm$  425.3 (96% of annual total) and 364.7  $\pm$  361.7 millions tons (87% of annual total) at Yichang and Datong, respectively, in the interval of 1950–2002. They decrease to 43.2  $\pm$  113.7 (98% of annual total) and 111.1  $\pm$  112.6 million tons (80% of annual total) in 2003–2015, respectively (Fig. 9). TGD sedimentation is the main reason of the dramatic reduction in the wet seasons (see discussion in Section 5.3).

## 4.3.4. Sediment rating curves

The sediment rating curves imply changes out of phase between the streamflow and sediment loads. We see that the lower CRI mainstream, i.e., at Datong, is overall much less sediment-loaded compared to Yichang and Cuntan (Fig. 10A). Over time, it becomes increasingly less sediment-loaded at all three stations in the mainstream due to strikingly reduced sediment loads. In the tributaries, the Han River and the three inlets are characterized by a higher sediment loading rate while the outflows of the Dongting and Poyang lakes are much less sediment-loaded. The annual sediment loads of the Han River and at the three inlets have decreased significantly considering streamflow reduction at a smaller rate, while the changes at Chenglingji and Hukou are comparatively less apparent because sedimentation and/or erosion inside the lakes have buffering impacts on sediment flux at the lake mouths. In all, these changes suggest that the CIR downstream of the TGD becomes much less sediment-loaded in the decades since the mid-1980s due to more remarkable reduction of sediment loads compared to streamflow.

# 5. Discussion

#### 5.1. Sensitivity to data range in inferring long-term changes

The length of a data series has considerable impacts on interpretation of hydrological changes. For instance, the significant change point of the annual streamflow at Yichang and Hankou in 1954 is detected based on data in 1865–2015 which is however not revealed by datasets in 1950–2015 (see Fig. 4B). Linear trend analysis of the streamflow at Datong shows a slight but nonsignificant decrease trend based on data in 1950–2015 whereas it exhibits a non-significant increase trend in 1955–2015. The later is more reasonable given a significant change point in 1954. The extreme values, e.g., much higher streamflow in 1954 and the late 1990s, influence the trend analysis results too.

The sediment loads changes at Datong provide another example of the sensitivity to data range. The time series of annual sediment loads at Datong do not show an apparent decrease or increase



Fig. 9. Intra-annual variations of monthly sediment loads at (A) Yichang and (B) Datong in the periods of 1950–2002 and 2003–2015. Only the mean values of monthly sediment discharges at Yichang are collected between 1950 and 2002 thus the variation range is not shown.



**Fig. 10.** The sediment rating curves (A) at Cuntan, Yichang, and Datong, and (B) at Chenglingji, Huangzhuang, Hukou, and three inlets based on yearly streamflow and sediment loads.

trend between 1950 and 1970, but a decrease trend is detected between the early 1960s and late 1990s, as well as a much faster decrease trend between 1980 and 2015 (see Figs. 8A and S10). These evidences explain why inconsistent interpretation was obtained by using data of different length in previous studies.

# 5.2. River-lake interplay and its hydrological impacts

The interactions between the CJR mainstream and the two lakes in terms of water and sediment exchanges play an essential role on the hydrological regime downstream of the TGD (Hu et al., 2007; Guo et al., 2011; Ou et al., 2012). Diversion of water and sediment from the mainstream to the Dongting Lake via the three inlets is only observed in the wet seasons when river stages in mainstream are higher than the bed level at the inlets. The diverted amount of water and sediment has reduced significantly because of lowered river stage in the mainstream and sedimentation in the inlet channels. The former is partly caused by bed erosion due to sediment starvation at the year-to-decade time scale and by decreased river discharges at the seasonal time scale. With non-significant change of total streamflow from the lake tributaries, the streamflow reduction at three inlets explains 89% of streamflow reduction at Chenglingji in 2002–2015 compared to 1951–1968 (see Section 4.2).

Interactions between mainstream and Poyang Lake have changed considerably as well. The Poyang Lake discharges water and sediment towards the mainstream and the mainstream has a 'backwater effect' on the lake outflows most of the time in a year course (Guo et al., 2011). Reverse flow happens in the wet seasons between July and September when the CIR is at a river stage higher than that in the Poyang Lake. River stage differences occur because of earlier start and ending of the rainy season in the middle-lower basin than that in the upper CJR basin as mentioned in Section 2.1. The magnitude and frequency of the reverse flow at Hukou however have decreased profoundly even vanished in the recent decade (Guo et al., 2011; Zhang et al., 2012b,c). In the mean time the annual streamflow at Hukou has changed little between 1950 and 2015 (see Fig. S5B). It is because the reverse flow is overall limited compared to the total streamflow of the lake tributaries while the later exhibits no directional changes.

The river-lake interactions are inherently controlled by river stages in the vicinity of river-lake conjunctions. There are highly linear correlationships between the river stages in the mainstream and at the lake mouth, such as that between Luoshan and Chenglingji and that between Jiujiang and Hukou (see Fig. S12). It implies that water level falling in the mainstream will induce synchronized water level drops at the lake mouth, thus lowering the 'backwater effect' (Guo et al., 2011; Mei et al., 2016). River stages along the mainstream fall because of seasonal river discharge decreases and channel degradation (Xu et al., 2011b; Dai and Liu, 2013). Channel degradation is evidenced by river bank erosion and bed scour (Xia et al., 2016) and by changed river dischargeriver stage rating curves. River stages are falling under the same river discharge at Yichang and Shashi, particularly under low and medium river discharges (see Fig. S13). For instance, the river stage under a river discharge of 10,000 m<sup>3</sup>/s drops from  $\sim$ 43.8 m in 1960, to ~42.7 m in 1990, ~42.1 in 2005, and ~41.3 m in 2015 at Yichang. Falling river stages in the mainstream cause consequent water level drop at the lake mouths, enlarging the water level gradients in the river-lake vicinity (Dai et al., 2015). The enlarged water level gradients foster faster lake outflow, which is in particular significant in the wet-to-dry transition season following TGD impoundments (see Fig. S14). It creates a 'draining effect' on the lake outflow, resulting in decreased water storage inside the lakes and lowered lake stage. The slightly increased river discharges along the mainstream in the dry seasons by the TGD may reinforce the backwater effects on the lake outflow but it is not able to restore water storage inside the lakes. As a consequence, low water occurrence is advanced in time, e.g., by  $\sim$ 20 days in the Dongting Lake (Fig. 11). Moreover, there are nonsignificant decreasing trends of precipitation (Zhang et al., 2011) and reference evapotranspiration (Ye et al., 2014) in autumn in the Poyang Lake basin. These changes augment the low water regime and can cause severe droughts in the dry seasons.

Lake size shrinkage in long-term further endangers water resources security in the lakes. The annually mean water levels at Chenglingii have increased between 1960s and 1990s but underwent a switch to decline since 2000s (Duan et al., 2012). Similarly, the mean water levels at Hukou and Xingzi, a stage station 38 km more inside the Poyang Lake compared to Hukou, have increased between 1960 and 2000 and switched to an afterward decline (Mei et al., 2016; Zhou et al., 2016). It is estimated that one meter water level drop will induce a surface area reduction and water storage volume shrinking by approximately 200–440 million m<sup>2</sup> and 1.3–1.9 billion m<sup>3</sup>, respectively, in the Dongting Lake according to the water level-lake area and water level-storage volume curves (Ding and Li, 2011). The yearly-averaged surface area of the Dongting Lake has decreased from 4350 km<sup>2</sup> in 1949 to 2518 km<sup>2</sup> in 1998 and 1837 km<sup>2</sup> in 2006 (Chen et al., 2001a; Feng et al., 2013; Zhao et al., 2013). The surface area of the Poyang Lake has decreased from 5160 km<sup>2</sup> in 1954 to 3860 km<sup>2</sup> in 1992 at a water level of 22 m at Hukou (Min, 2000). In fact the water level at Hukou rarely exceeds 22 m in the recent decade when the yearlyaveraged lake area is only 1740 ± 520 km<sup>2</sup> (Feng et al., 2012). Lake sedimentation and extensive reclamation of the low-lying lands are the main causes of the lake shrinkage. On the other hand, extensive sand mining inside the lakes (Lai et al., 2014) and a sediment balance shift from lake sedimentation to erosion (see Section 4.3) may slow down the shrink trend to some degree. Anyhow reduced lake area and storage volume at a decadal time scale implies degenerated function of the lakes in buffering intraannual river discharge changes along the mainstream.

Lake size reduction, seasonal river discharge decline and associated draining effects enhance low water occurrence and severity inside the lakes. Given that it is dry season following TGD impoundments, hastened and persistent low water conditions impair water security and riparian ecosystem in the lakes. The wetlands in the Dongting Lake and Poyang Lake, as valuable habitats for fishes and migratory birds in the eastern Asia, are exposed for longer time, particularly in the dry seasons between October and March (Xu et al., 2013b: Sun et al., 2012; Mei et al., 2016). In October 2006 and May 2011, for instance, the wet surface area of the Poyang Lake is estimated by merely  $\sim$ 50 km<sup>2</sup> due to persistent low water level (Zhang et al., 2012b; Feng et al., 2013). As a result, it causes enormous growth of pioneer species such as poplar and over-reproduction of Microtus fortis etc., endangering lake ecosystem as valuable natural reserves (Chang et al., 2010; Sun et al., 2012; Xu et al., 2013b). To protect the lakes from drying, measures such as constructing sluices at the lake mouths to control lake outflow are under debating (Wang et al., 2015). But it remains insufficiently known how such a sluice will change the river-lake interactions further and has impacts on sediment imbalance, flooding risk, and lake water quality etc., and integrated research work is needed to sort out nature-based solutions.

The river–lake interplay also plays an important role in mitigating flood risk in the middle-lower CJR basin. The TGD can proactively reduce outflow when its inflow is >55,000 m<sup>3</sup>/s, thus protecting the downstream basin, particularly the reaches between



Fig. 11. Flow charts of the hydrological regime changes and the consequences downstream of the TGD in the CJR basin.

Yichang and Luoshan, from big floods formed in the upper basin. For instance, the inflow of TGD was as large as  $70,000 \text{ m}^3/\text{s}$  in July of 2010 and 2012 while the outflow was reduced to  $<50,000 \text{ m}^3/\text{s}$  (see Fig. S2). In addition, the TGD can reactively reduce outflow when the downstream basin encounters floods, thus lowering mainstream river stage and providing beneficial conditions for downstream floods flushing. By that the TGD is able to prevent occurrence of 1/100 probability disastrous floods as that in 1954 and 1998 in the downstream basin. Even though the TGD can not regulate regional river floods within the lake systems caused by local strong rain storms. The effects of TGD in mitigating basin-scale big floods need to be evaluated in long-term and shall not be underestimated and overestimated.

The above discussion suggests that it is not only the magnitude of river discharge but also its frequency, timing, duration, and rate of change that matters in basin-scale hydrological processes. As the TGD may be obligated to keep the pool level as high, as early, and as long as possible for more hydropower generation by advancing impoundment and slowing flushing (Li et al., 2013b), integrated management considering flood control, power generation, outflow reduction, and resultant influences on the downstream basin is of pivotal importance. The successive droughts occurred in recent years remind that low water management is also badly needed other than flood risk mitigation.

# 5.3. Causes of the hydrological changes

# 5.3.1. Causes of streamflow changes

Natural hydro-meteorological changes and human activities are generally two main forces driving hydrological changes at river basin scales. We see that the basin-wide air temperature has increased since the mid-1980s (Chen et al., 2001a; Sang et al., 2012), and precipitation has decreased since 1950s and evaporation rate is decreasing since 1960s (Wang et al., 2011). At century time scales, the actual evapotranspiration and the ratio of actual evapotranspiration to precipitation has increased in 1955-2008 compared to 1901–1954 in the CJR basin, i.e., by 30 mm per year and 0.06 per year, respectively (Jaramillo and Destouni, 2015). which may explain the detected change point in 1954 and the afterward streamflow reduction (see Figs. 4 and S4). There are attempts to isolate the impacts of natural climate changes and human interferences on the streamflow variations. For instance, there is a well established correlationship between basin-wide averaged precipitation and streamflow at Datong, suggesting a dominant role of precipitation on streamflow changes (Yang et al., 2005; Jiang et al., 2007; Zhao et al., 2017). Similarly, Wang et al. (2013) inferred that rainfall variations may explain 29% of streamflow changes in the CJR basin between 1970 and 2008. Note that there is large variability in the relationship between precipitation and streamflow, thus consideration of both precipitation and evapotranspiration is necessary to isolate meteorological controls on streamflow changes.

Human interferences play an increasingly significant role in modulating the hydrological regime. Annual water consumption (consumptive use of water resources for agriculture and industry and domestic purposes) has increased from 15 to 90 billion m<sup>3</sup> between 1949 and 2000 due to booming economy and population within the CJR basin (MWR, 1997–2005; Xu et al., 2007). The ongoing South-to-North water transfer project withdraws annually 18.3 billion m<sup>3</sup> of surface water out of the CJR basin to the north China since 2014 and the total diversion capacity will be 44.8 billion m<sup>3</sup> per year after completion of the project in the near future (OSNWDPCC, 2016). The number of hydropower dams has increased exponentially since 1950s, reaching >50,000 nowadays (Yang et al., 2011). The total reservoir storage capacity has increased markedly, reaching 200 billion m<sup>3</sup> by 2003 (Yang et al.,

2004) with 55.69 billion m<sup>3</sup> in the upper basin by 2005 (Wei et al., 2011). The dams recently constructed and under constructions in the upper basin are not yet included. Increasing water consumption and water withdrawal causes net reduction of streamflow while the dams mainly modulate the timing and magnitude of river discharges. We see that the seasonal streamflow changes are more profound compared to the inter-annual changes, and human activities are ascribed to be the main cause of that.

## 5.3.2. Causes of sediment loads changes

Sediment loads changes are very much influenced by human activities. Rainfall variations affect sediment yield through surface erosion (sediment source) while the later is also influenced by human-related land use changes. Sediment loads reduction is much more profound than streamflow changes (see Fig. 10), suggesting that sediment transport is source-limited instead of transport-limited. It was reported that the land surface area subject to hydraulic erosion nearly doubled between 1950 and 2001 over the entire CJR basin, i.e., from 364 to 707 billion m<sup>2</sup> (Yang et al., 2004), suggesting possibly increasing sediment yield. On the other hand, implementation of anti-deforestation measures in the upper CJR basin since 1980s reduces the area and intensity of surface erosion (Yang et al., 2006; Hu et al., 2009). Forest coverage rate is estimated by an increase from 33.8% to 56.2% in the upper basin by 2005 (Wei et al., 2011). Analysis of satellite images in 1980s and 2010, however, shows only marginally increased forest area and residential land area and decreased irrigated land and dry farmland areas within the CJR basin, and the change rates are overall small (<4%) (Zhao et al., 2017). These studies are somehow inconsistent though reduced sediment yield in the upper basin due to land use changes is widely recognized as one of the main causes of sediment loads reduction at Cuntan since the mid-1980s (Yang et al., 2006, 2011; Wang et al., 2007b; Hu et al., 2009).

Reservoir sedimentation is another main reason of sediment loads reduction (Yang et al., 2006; Wang et al., 2007b). Yang et al. (2006) suggested that reservoirs were responsible for 66% of sediment loads reduction at Cuntan and Yichang since the mid-1980s. Downstream of Cuntan. the TGD retains annually 125.7 million tons sediment between 2003 and 2015, which is averagely 76% of its sediment influx (Table 2). A major portion of the reservoir sedimentation (>95%) occurs in the wet seasons (Xu and Milliman, 2009; CWRM, 2006, 2016), explaining pronounced sediment loads reduction in the wet seasons (see Fig. 9). By adding the sediment trapped by the TGD to the measured sediment loads at Yichang and Datong, we see that the linear reduction trend dating back to the early 1980s and mid-1960s at the two stations, respectively, continues until nowadays (see Fig. S10). We also estimate that the sediment loads reduction at Yichang and Datong in the period of 2003-2015 would be 76% and 49% smaller, respectively, if there is no TGD (see Table 2 and Fig. S10). The newly constructed big dams in the upper mainstream, e.g., the Xiangjiaba Dam, have caused additional substantial sediment loads decrease at Cuntan since 2012. It will slow down TGD sedimentation and in the mean time further reduce sediment loads downstream of TGD (Table 2).

Other human activities having impacts on sediment loads include river regulation, lake sedimentation, and sand mining and associated river morphological changes. The sharp reduction of sediment loads at the three inlets in the late-1960s was ascribed to river regulations in the Jing River (the meander river between Yichang and Luoshan), i.e., meander cutoff and river length reduction (Chen et al., 2010). Sand mining (i.e., sediment extraction from the river bed) in the CJR mainstream is overall limited (Dai and Liu, 2013), but intensive sand mining within the Poyang Lake (Lai et al., 2014), by stirring the lake bed and enhancing suspension transport, causes notably increased sediment loads at Hukou (see Fig. S11C). The Dongting Lake changes from a sediment sink to a source

Year/ Period	TGR In <sup>°</sup>	TGR Out <sup>**</sup>	TGR sedimentation	Datong
1956-2002	448.7	488.7	-40.0	424.5
2003	220.4	84.0	136.4	206.0
2004	183.8	63.7	120.1	147.0
2005	274.4	103.0	171.4	216.0
2006	112.4	8.9	103.5	84.8
2007	220.4	50.9	169.5	138.0
2008	216.9	32.2	184.7	130.0
2009	174.4	36.0	138.4	111.0
2010	216.6	32.8	183.8	185.0
2011	93.1	6.9	86.2	71.8
2012	219.0	45.3	173.7	161.0
2013	126.8	32.8	94.0	117.0
2014	55.4	10.5	44.9	120.0
2015	32.0	4.2	27.8	116.0
2016	42.2	8.8	33.4	152.0
2003-2015	165.0	39.3	125.7	138.7

 Table 2

 Annual sediment inflows and outflows (million tons) of the TGD and the sediment loads at Datong. Data are from CWRM (2006, 2016).

regarding sediment exchange with the CJR mainstream since 2000 (see Fig. S11A). Smaller sediment loads at Datong compared to the summed sediment loads at Yichang and that from the three main tributaries suggest river aggradations in 1978-2001 (see Figs. S10 and S11). There is a switch from river deposition to erosion since 2002 which is evidenced by morphological survey (Li et al., 2009; CWRM, 2016). For instance, an erosion of 1001.3 million m<sup>3</sup> is observed along the river between Yichang and Luoshan (a river length of 408 km) between October 2002 and October 2015 (CWRM, 2016), and the erosion volume is  $\sim$ 201.1 million m<sup>3</sup> in the river between Luoshan and Hankou (a river length of 251 km) between November 2003 and November 2015 (CWRM, 2016). Considering a bulk density of 1.05 tons per m<sup>3</sup>, the sediment erosion rate is ~105.2 million tons per year in the river segment between Yichang and Hankou. It suggests that river erosion in the mainstream and the sediment source from main tributaries contribute  $\sim$ 70% and  $\sim$ 20% of sediment loads at Datong, respectively, in 2003–2015. It is these sediment sources that mitigating sediment loads reduction at Hankou and Datong compared to Yichang (see Figs. 8 and S10).

So far we see that the TGD's impacts on the hydrological regime are not confined to river discharge regulation and sediment trapping but also include indirect influences in terms of reinforcing downstream river erosion and altering river–lake interplay (Fig. 11). The TGD's impacts on river discharges are more profound at seasonal time scales while its impacts on sediment loads reduction are more pronounced at annual to decadal time scales. Overall we see that climatic changes and human activities in the upper basin have caused remarkable river discharge and sediment loads changes, and the TGD exerts additional modification on that and causes subsequent changes in the downstream river while alongriver buffering dissipate the changes in the downward direction.

# 5.4. Implications of the hydrological changes

The changed hydrological regime has broad impacts and consequences on the river morphology and ecosystem and other relevant river functions. The CJR downstream of Yichang is a 'golden waterway' regarding its importance on water transportation. Its navigation channel depth is threatened by the seasonally lowered river stages. The suspended sediment concentrations have reduced from 0.467  $\pm$  0.539 kg/m<sup>3</sup> under the pre-TGD condition to 0.070  $\pm$ 0.108 kg/m<sup>3</sup> under the post-TGD condition at Yichang (Xiao and Duan, 2011), which can have profound impacts on the aquatic ecosystem given increased water transparency and light penetration. Downstream channel degradation in terms of bed scour and bank erosion and collapse threatens the safety of the dikes and other flood defense structures (Li et al., 2009; Xia et al., 2016). As a result of bed erosion, bed sediment is coarsening, i.e., changes from coarse sand ( $\sim$ mm) before 2003 to gravel and pebble ( $\sim$ cm) in 2008 at Yichang (Luo et al., 2012; Yang et al., 2018), which will stimulate downward development of bed scour until a new river morphological equilibrium state is reached in long-term (decades to a century). The altered river-lake interplay and shrinking lakes will lower flood resilience in the middle-lower CJR basin due to decreased lake storage capacity (Li et al., 2009; Fang et al., 2012), and increase drought occurrence probability and impair lake ecosystem. Moreover, dramatically declined sediment loads are accompanied by reduced nutrient loading because the suspended sediments are composed of dominantly fine material (e.g., median size of  $\sim 10 \,\mu\text{m}$  at Datong) and are transported in the form of aggregated flocs (Guo and He, 2011). For instance, there is a 77% reduction of total phosphorous and an 83% reduction of particulate phosphorous annually (Zhou et al., 2013) and also a substantial decrease of silica flux (Ran et al., 2016). These changes would have impacts on the biogeochemical cycling in the river-estuary-coastal ocean continuum (Zhang et al., 2015). These issues deserve attention from both research and management point of view.

The changed river and sediment discharges impose a threat on the delta with respect to salt intrusion and delta erosion. River discharge provides a critical force in inhibiting salt intrusion driven by marine tides and waves in the delta region. Landward salt intrusion affects freshwater intake in the CJR delta in the dry seasons when the river discharge at Datong is  $<15,000 \text{ m}^3/\text{s}$ , threatening freshwater supply to millions of people in Shanghai (Luo and Shen, 1994). Modeling studies have showed that TGD impoundments in autumn tend to advance the timing and enhance the intensity of salt intrusion while larger river discharges in the dry seasons may relieve salt intrusion to some degree (Qiu and Zhu, 2013). In addition, water withdrawal along the river downstream of Datong is considerable and will further reduce the river discharge in counteracting salt intrusion (Zhang et al., 2012a). The altered river discharge hydrograph is highly likely to hasten the onset of and induce more landward saltwater intrusion in the CJR delta in the wet-to-dry transition seasons. Regulating lake outflow by constructing sluices at the lake mouths may worsen this situation. Maintaining a river discharge at Datong larger than a minimum threshold (e.g., 12,000 m<sup>3</sup>/s) through optimized dam operation can help to reduce the salt intrusion risk.

Delta erosion and salt marshes loss is a global concern and also a big issue in the CJR delta. Yang et al. (2011) had reported a shift from accretion to erosion around the sub-merged CJR delta (the study area is selective and small compared to the entire CJR delta) in the recent decade and ascribed it to reduced sediment supply. On the other hand, the CJR delta as whole has not shown directional erosion in the period of 1958–2010 (Luan et al., 2016; Zhao et al., 2018). So far it is widely known that the deposition rate in the CJR delta has decreased measurably in the recent decade though overwhelming erosion has not emerged, which can be ascribed to a time lag effect between deltaic morphological response and sediment supply reduction. It is reasonably expected that the sediment loads at Datong will remain low in the coming decades to century, thus delta erosion is somehow inevitable in long term. Delta safety and ecosystem is at high risk considering sediment loads reduction, sea-level rise and land subsidence. Research work to figure out the effects of sediment redistribution within the delta in causing a time lag of large scale deltaic morphological adaptation is needed.

# 6. Conclusions

In this work we collect and examine a long time series of river discharges and sediment loads data in the CJR basin downstream of the TGD and elaborate the hydrological regime changes at seasonal, annual and decadal time scales. The causes and implications of those changes are discussed and the role of the TGD is evaluated.

The annual streamflow changes are non-significant downstream of TGD other than a significant change point in 1954. Streamflow changes are more remarkable at seasonal time scales, such as decreased flood peak and autumn discharges and increased dry season discharges. The hydro-climate changes occurred in the upper basin and the regulation effects of TGD together explain the seasonal changes.

Sediment loads have decreased progressively in the CJR basin. The decrease trend dates back to the mid-1980s and the TGD accelerates the decrease since 2003. The sediment loads immediately below TGD have decreased by 92% in 2003–2015 compared to 1950–1985 at Yichang. The TGD is responsible for 76% and 49% of the sediment loads reduction at Yichang and Datong, respectively. As a result, subsequent bed scour and bank erosion occur in the downstream river. The eroded sediment, together with the sediment loads from the tributaries, mitigates sediment reduction in the down river direction.

The altered river–lake interplay in response to channel degradation and seasonal river discharge reduction is highly responsible for low stages in the lakes. Substantial river stage falling downstream of the TGD and enlarged water level gradients in the vicinities of river–lake conjunctions induce a 'draining effect' fostering faster lake outflow following TGD impoundments. It results in lower stage and reduced lake water storage which threatens water resources security and riparian ecosystem in the lakes.

The hydrological changes have a wide scope of influences and consequences on water security, flooding and drought risk management, and delta safety. More big dams are under construction along the mainstream upward TGD and they will further regulate the river discharge and sediment delivery processes. We identify knowledge gaps in research, including: (1) there lacks of indepth analysis of climatic and hydrological changes and their coherent behavior at different time and space scales, in particular regarding the role of nature and human activities (e.g., land use changes, big hydropower dams, water transfer and consumption) on streamflow changes based on analysis of precipitation, evapotranspiration, and streamflow data together; (2) insufficient understanding of the climatic impacts of the series of large reservoirs including the TGD and the others under construction in the upper basin; (3) integrated water resources management and flooding and drought hazard management when considering both surface water and groundwater, gross water storage changes, optimized hydropower dam operations, and fluvial ecosystem restoration opportunity etc., and last but not least (4) study of basin-scale hydrological and geomorphologic changes and ecosystem evolution given a doctrine switch from 'economical development first' to 'environment protection and ecosystem conservation first'. Coping with these issues will need comprehensive management strategies by taking the entire river basin into consideration regarding water, sediment, and associated resources, ecology, and safety.

### Acknowledgements

This study was funded by National Science Foundation of China (Nos. 51320105005; 51739005; 41506105; 41776062) and Shanghai Committee of Science and Technology (Nos. 17DZ1204800; 16DZ1205403). L. Guo was also supported by SKLEC-fund (No. 2015RCDW02) and China Post-Doc Fund (No. 2016T90351). The data used in this study are available from official websites of the Hydro-information Center of the Ministry of Water Resources of China (http://xxfb.hydroinfo.gov.cn), the China Three Gorges Corporation (http://www.ctg.com.cn/sxjt/sqgk/index.html), and open-accessed bulletins published by Ministry of Water Resources of China (http://www.mwr.gov.cn/sj/tjgb/zghlnsgb/). Special thanks are given to Li Yan (Xiamen University), Lv Houyuan (Academy of China Science), Shen Jian (VIMS) for their internal review and constructive comments. Constructive comments from two anomalous reviewers and the editor help to improve this work enormously thus are very much appreciated.

# Appendix A. Supplementary data

Supplementary data associated with this article can be found, in the online version, at https://doi.org/10.1016/j.jhydrol.2018.03. 035.

#### References

- Arnell, N.W., Gosling, S.N., 2013. The impacts of climate change on river flow regimes at the global scale. J. Hydrol. 486, 351–364.
- Becker, S., Gemmer, M., Jiang, T., 2006. Spatiotemporal analysis of precipitation trends in the Yangtze River catchment. Stoch. Env. Res. Risk Assess. 20 (6), 435– 444.
- Birkinshaw, S.J., Guerreiro, S.B., Nicholson, A., Liang, Q.H., Quinn, P., Zhang, L.L., He, B., Yin, J.X., Fowler, H.J., 2017. Climate change impacts on the Yangtze River discharge at the Three Gorges Dam. Hydrol. Earth Syst. Sci. 21, 1911–1927.
- Brandt, S.A., 2000. Classification of geomorphological effects downstream of dams. CATENA 40, 375–401.
- Castino, F., Bookhagen, B., Strecker, M.R., 2016. River-discharge dynamics in the Southern Central Andes and the 1976–1977 global climate shift. Geophys. Res. Lett. 43. https://doi.org/10.1002/2016GL70868.
- Chang, J., Li, J.B., Lu, D.Q., Zhu, X., Lu, C.Z., Zhou, Y.Y., Deng, C.X., 2010. The hydrological effect between Jingjiang River and Dongting Lake during the initial period of Three Gorges Project operation. J. Geog. Sci. 20 (5), 771–786.
- Chao, B.F., Wu, Y.H., Li, Y.S., 2008. Impacts of artificial reservoir water impoundment on global sea level. Science 320, 212–214.
- Chen, J., Finlayson, B.L., Wei, T.Y., Sun, Q.L., Webber, M., Li, M.T., Chen, Z.Y., 2016. Changes in monthly flows in the Yangtze River, China- with special reference to the Three Gorges Dam. J. Hydrol. 536, 293–301.
- Chen, X.Q., Zong, Y.Q., Zhang, E.F., Xu, J.G., Li, S.J., 2001a. Human impact on the Changjiang (Yangtze) River basin, China, with special reference to the impacts on the dry season water discharge into the sea. Geomorphology 41, 111–123.
- Chen, Z.Y., Li, J.F., Shen, H.T., Wang, Z.H., 2001b. Yangtze River of China: historical analysis of discharge variability and sediment flux. Geomorphology 41, 77–91.
- Chen, Z.Y., Wang, Z.H., Finlayson, B., Chen, J., Yin, D.W., 2010. Implications of flow control by the Three Gorges Dam on sediment and channel dynamics of the middle Yangtze (Changjiang) River, China. Geology 38 (11), 1043–1046.
- China Water Resources Ministry (CWRM) (ed.). 2006. River sediment bulletin of China 2005. Water Publication Press, Beijing, China, pp.1–17.
- China Water Resources Ministry (CWRM) (ed.). 2016. River sediment bulletin of China 2015. Water Publication Press, Beijing, China, pp.1–23.
- Dai, S.B., Lu, X.X., 2013. Sediment load change in the Yangtze River (Changjiang): a review. Geomorphology. https://doi.org/10.1016/j.geomorph.2013.05.027.
- Dai, X., Wang, R.R., Yang, G.S., 2015. Non-stationary water-level fluctuation in China's Poyang Lake and its interactions with Yangtze River. J. Geog. Sci. 25 (3), 274–288.

- Dai, Z.L. Liu, I.T., 2013. Impacts of large dams on downstream fluvial sedimentation: an example of the Three Gorges Dam (TGD) on the Changjiang (Yangtze River). J. Hydrol. 480, 10-18.
- Dai, Z.J., Chu, A., Stive, M.J.F., Li, J.F., 2011. Is the Three Georges Dam the cause behind the 2006 extreme low suspended sediment discharge into the Yangtze (Changjiang) estuary? Hydrol. Sci. J. 56 (7), 1280-1288.
- Degu, A.M., Hossain, F., Niyogi, D., Pielke, R., Shepherd, J.M., Voisin, N., Chronis, T., 2011. The influence of large dams on surrounding climate and precipitation Geophys. Res. Lett. 38, L04405. https://doi.org/10.1029/ patterns. 2010GL046482.
- Deng, H.Q., Luo, Y., Yao, Y., Liu, C., 2013. Spring and summer precipitation changes from 1880 to 2011 and the future projections from CMIP5 models in the Yangtze River Basin, China. Quat. Int. 304, 95-106.
- Ding, X.W., Li, X.F., 2011. Monitoring of the water-area variations of Lake Dongting in China with ENVISAT ASAR images. Int. J. Appl. Earth Obs. Geoinf. 13, 894-901.
- Duan, K., Xiao, W.H., Mei, Y.D., Hao, C.L., Wang, Y., 2012. An analysis of the impacts of large water projects on water resources utilization in the Dongting Lake regions. Resour. Environ. Yangtze Basin 21 (11), 1389-1392 (in Chinese with English abstract).
- Fang, C.M., Dong, Y.H. (Eds.), 2011. Sedimentation in the Three Gorges Reservoir and its Impacts and Management Strategies. Changjiang Press, Wuhan, China.
- Fang, H.W., Han, D., He, G., Chen, M., 2012. Flood management selections for the Yangtze River midstream after the Three Gorges Project operation. J. Hydrol. 432-433, 1-11.
- Fekete, B.M., Vörösmarty, C.J., Grabs, W., 2002. High-resolution fields of global runoff combining observed river discharge and simulated water balances. Global Biogeochem. Cycles 16 (3), 1042. https://doi.org/10.1029/ 1999GB001254.
- Feng, L., Hu, C.M., Chen, X.L., Cai, X.B., Tian, L.Q., Gan, W.X., 2012. Assessment of inundation changes of the Poyang Lake using MODIS observations between 2000 and 2010. Remote Sens. Environ. 121, 80-92.
- Feng, L., Hu, C.M., Chen, X.L., Zhao, X., 2013. Dramatic inundation changes of China's two largest freshwater lakes linked to the Three Gorges Dam. Environ. Sci. Technol. 47, 9628-9634.
- Gao, B., Yang, D.W., Yang, H.B., 2013. Impact of the Three Gorges Dam on flow regime in the middle and lower Yangtze River. Quat. Int. 304, 43-50.
- Gemmer, M., Jiang, T., Su, B.D., Kundzewicz, Z.W., 2008. Seasonal precipitation changes in the wet season and their influence on flood/drought hazards in the Yangtze River basin, China. Quat. Int. 186 (1), 12–21.
- Graf, W.L., 2006. Downstream hydrologic and geomorphic effects of large dams on American rivers. Geomorphology 79, 336-360.
- Guo, H., Hu, Q., Zhang, Q., Feng, S., 2011. Effects of the Three Gorges Dam on the Yangtze River flow and river interaction with Poyang Lake, China: 2003–2008. J. Hydrol. 416-417, 19-27.
- Guo, L.C., He, Q., 2011. Freshwater flocculation of suspended sediments in the Yangtze River, China. Ocean Dyn. 61 (2-3), 371-386.
- Hu, B.Q., Yang, Z.S., Wang, H.J., Sun, X.X., Bi, N.S., Li, G.G., 2009. Sedimentation in the Three Gorges Dam and the future trend of Changjiang (Yangtze River) sediment flux to the sea. Hydrol. Earth Syst. Sci. 13, 2253-2264.
- Hu, Q., Feng, S., Guo, H., Chen, G.Y., Jian, T., 2007. Interactions of the Yangtze River flow and hydrological processes of the Poyang Lake, China. Journal of Hydrology 347, 90–100.
- IPCC, 2013. Summary for Policymakers. In: Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Stocker, T.F., D. Qin, G.-K. Plattner, M. Tignor, S.K. Allen, J. Boschung, A. Nauels, Y. Xia, V. Bex and P.M. Midgley (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York NY USA
- Jaramillo, F., Destouni, G., 2015. Local flow regulation and irrigation raise global human water consumption and footprint. Science 350, 1248–1251.
- Jiang, T., Su, B.D., Hartmann, H., 2007. Temporal and spatial trends of precipitation and river flow in the Yangtze River basin, 1961–2000. Geomorphology 85, 143– 154
- Kendall, M.G., 1975. Rank Correlation Methods. Griffin, London.
- Kuang, X.Y., Zhang, Y.C., Huang, Y., Huang, D.Q., 2014. Changes in the frequencies of record-breaking temperature events in China and its association with East Asian Winter Monsoon variability. J. Geophys. Res.: Atmos. 119. https://doi.org/ 10.1002/2013ID020965.
- Lai, X.J., Shankman, D., Huber, C., Yesou, H., Huang, Q., Jiang, J.H., 2014. Sand mining and increasing Poyang Lake's discharge ability: a reassessment of causes for lake decline in China. J. Hydrol. 519, 1698–1706.
- Li, K.F., Zhu, C., Wu, L., Huang, L.Y., 2013a. Problems caused by the Three Gorges Dam construction in the Yangtze River basin: a review. Environ. Rev. 21 (3), 127-135
- Li, N., Wang, L.C., Zeng, C.F., Wang, D., Liu, D.F., Wu, X.T., 2016. Variations of runoff and sediment load in the middle and lower reaches of the Yangtze River, China (1950-2013) pp. 160-154 PLoS One 11 (8). https://doi.org/10.1371/journal. pone.0160154
- Li, S., Xiong, L.H., Dong, L.H., Zhang, J., 2012. Effects of the Three Gorges Reservoir on the hydrological droughts at the downstream Yichang station during 2003-2011. Hydrol. Process. 27 (26), 3981-3993.
- Li, Y., Guo, S.L., Guo, H.J., Zhang, H.G., Ding, S.X., 2013b. Flood control risk and benefit of impounding water in advance for the Three Gorges Reservoir. J. Yangtze River Sci. Res. Inst. 30 (1), 8-14.

- Li, Y.T., Sun, Z.H., Liu, Y., Deng, J.Y., 2009. Channel degradation downstream from the Three Gorges Project and its impact on flood level. J. Hydraul. Eng. 135, 718-728
- Luan, H.L., Ding, P.X., Wang, Z.B., Ge, J.Z., Yang, S.L., 2016. Decadal morphological evolution of the Yangtze Estuary in response to river input changes and estuarine engineering projects. Geomorphology 265, 12-23.
- Luo, B.Z., Shen, H.T., 1994. The Three Gorges Project and the Ecological Environment. Science Press, Beijing, China, pp.50-53 (in Chinese).
- Luo, X.X., Yang, S.L., Zhang, J., 2012. The impact of the Three Gorges Dam on the downstream distribution and texture of sediments along the middle and lower Yangtze River and its estuary, and subsequent sediment dispersal in the East China Sea. Geomorphology 179, 126-140.
- Mann, H.B., 1945. Nonparametric tests against trend. Econometrica 13, 245-259.
- Mei, X.F., Dai, Z.J., Fagherazzi, S., Chen, J.Y., 2016. Dramatic variations in emergent wetland area in China's largest freshwater lake, Poyang Lake. Adv. Water Resour. https://doi.org/10.1016/j.advwatres.2016.06.003.
- Miller, N.L., Jin, J.M., Tsang, C.-F., 2005. Local climate sensitivity of the Three Gorges Dam. Geophys. Res. Lett. 32, L16704. https://doi.org/10.1029/2005GL022821.
- Milliman, J.D., 1997. Blessed dams or damned dams. Nature 386, 325-326.
- Min, Q., 2000. Study on the relationship between shape, water regime and innings of the Poyang Lake. Adv. Water Sci. 11 (1), 76-81 (in Chinese with an abstract in English)
- Ministry of Water Resources of the People's Republic of China (MWR), 1997-2005. Bulletin of Water Resources in China. http://www.mwr.gov.cn/zwzc/hygb/ szygb (in Chinese).
- Nakaegawa, T., Kitoh, A., Hosaka, M., 2013. Discharge of major global rivers in the 21st century climate projected with the high horizontal resolution MRI-AGCSMs. Hydrol. Process. 27, 3301-3318.
- National Climate Center of China Meteorological Administration (NCC), 2011. Analysis and evaluation of the climatic impact of the Three Gorges Project. China Three Gorges 9, 28-30 (in Chinese).
- Nilsson, C., Reidy, C.A., Dynesius, M., Revenga, C., 2005. Fragmentation and flow regulation of the world's largest river systems. Science 308, 405-408.
- Ou, C.M., Li, J.B., Zhang, Z.Q., Li, X.C., Yu, G., Liao, X.H., 2012. Effects of the dispatch modes of the Three Gorges Reservoir on the water regimes in the Dongting Lake area in typical years. J. Geog. Sci. 22, 594-608.
- Office of the South-to-North Water Diversion Project Construction Committee (OSNWDPCC), 2016. The South-to-North water diversion project. Engineering 2, 265-267.
- Qiu, C., Zhu, J.R., 2013. Influence of seasonal runoff regulation by the Three Gorges Reservoir on saltwater intrusion in the Changjiang River estuary. Cont. Shelf Res. 71, 16-26.
- Ran, X.B., Liu, S., Liu, J., Zhang, J.Y., Che, H., Ma, Y.X., Wang, Y.B., 2016. Composition and variability in the export of biogenic silica in the Changjiang River and the effect of the Three Gorges Reservoir. Sci. Total Environ. 571, 1191–1199.
- Sang, Y.F., Wang, Z., Liu, C., 2012. Spatial and temporal variability of daily temperature during 1961-2010 in the Yangtze River Basin, China. Quat. Int. https://doi.org/10.1016/j.quaint.2012.05.026.
- Sun, Z.D., Huang, O., Opp, C., Hennig, T., Marold, U., 2012. Impacts and implication of major changes caused by the Three Gorges Dam in the middle reaches of the Yangtze River, China. Water Resour. Manage. 26, 3367–3378.
- Sun, J.L., Lei, X.H., Tian, Y., Liao, W.H., Wang, Y.H., 2013. Hydrological impacts of climate change in the upper reaches of the Yangtze River basin. Quat. Int. 304, 62-74.
- Syvitski, J.M., Vörösmarty, C.J., Kettner, A.J., Green, P., 2005. Impact of humans on the flux of terrestrial sediment to the global coastal ocean. Science 308, 376-380.
- Taylor, W.A., 2000. Change-point analysis: a powerful new tool for detecting changes. Analysis, 1-19.
- Walling, D.E., Fang, D., 2003. Recent trends in the suspended sediment loads of the world's rivers. Global Planet. Chang 39, 111-126.
- Wang, H.J., Yang, Z.S., Wang, Y., Saito, Y., Liu, J.P., 2008. Reconstruction of sediment flux from the Changjiang (Yangtze River) to the sea since the 1860s. J. Hydrol. 349 (3), 318-332.
- Wang, P., Lai, G.Y., Li, L., 2015. Predicting the hydrological impacts of the Poyang Lake Project using an EFDC model. J. Hydrol. Eng. 21 (12). https://doi.org/ 10.1061/(ASCE)HE.1943-5584.0001240.
- Wang, W.G., Xing, W.Q., Yang, T., Shao, Q.X., Peng, S.Z., Yu, Z.B., Yong, B., 2012. Characterizing the changing behaviors of precipitation concentration in the Yangtze River Basin, China. Hydrological Processes 27 (24), 3375-3393.
- Wang, Y., Ding, Y.J., Ye, B.S., Liu, F.J., Wang, J., Wang, J., 2013. Contributions of climate and human activities to changes in runoff of the Yellow and Yangtze rivers from 1950 to 2008. Sci. China: Earth Sci. 43 (7), 1207-1219 (in Chinese).
- Wang, Y., Jiang, T., Bothe, O., Fraedrich, K., 2007a. Changes of pan evaporation and reference evapotranspiration in the Yangtze River basin. Theor. Appl. Climatol. 90.13-23.
- Wang, Y.J., Liu, B., Su, B.D., Zhai, J.Q., Gemmer, M., 2011. Trends of calculated and simulated actual evaporation in the Yangtze River basin. J. Clim. 24, 4494–4507. Wang, Z.Y., Li, Y.T., He, P.Y., 2007b. Sediment budget of the Yangtze River. Water
- Resour. Res. 43, W04401. https://doi.org/10.1029/2006WR005012.
- Wei, J., He, X.B., Bao, Y.H., 2011. Anthropogenic impacts on suspended sediment load in the upper Yangtze River. Reg. Environ. Change 11, 857-868.
- Wei, W., Chang, Y.P., Dai, Z.J., 2013. Streamflow changes of the Changjiang (Yangtze) River in the recent 60 years: Impacts of the East Asian summer monsoon, ENSO, and human activities. Quat. Int. 336, 98-107.

Woodward, J.C., Macklin, M.G., Krom, M.D., Williams, M.A.J., 2007. The Nile: evolution, quaternary river environments and material fluxes. In: Gupta, A. (Ed.), Large rivers: geomorphology and management. Wiley, Chichester, pp. 261–292.

- Word Meteorological Organization (WMO), 2016. WMO statement on the status of global climate in 2015. WMO-No. 1167. WMO, Genewa, Switzerland.
- Wu, J., Gao, X.J., Giorgi, F., Chen, Z.H., Yu, D.F., 2012. Climate effects of the Three Gorges Reservoir as simulated by a high resolution double nested regional climate model. Quat. Int. 282, 27–36.
- Xia, J.Q., Deng, S.S., Lu, J.Y., Xu, Q.X., Zong, Q.L., Tan, G.M., 2016. Dynamic channel adjustment in the Jingjiang Reach of the middle Yangtze River. Sci. Rep. https:// doi.org/10.1038/srep22802.
- Xiao, H., Duan, Z.H., 2011. Hydrological and water chemical factors in the Yichang reach of the Yangtze River pre- and post-impoundment of the Three Gorges Reservoir: consequences for the Chinese sturgeon Acipenser sinensis spawning population (a perspective). J. Appl. Ichthyol. 27, 387–393.
- Xing, W.Q., Wang, W.G., Shao, Q.X., Yu, Z.B., Yang, T., Fu, J.Y., 2016. Periodic fluctuation of reference evapotranspiration during the past five decades: does evaporation paradox really exist in China? Sci. Rep. 6, 39503. https://doi.org/ 10.1038/srep39503.
- Xu, C.Y., Singh, V.P., 2004. Review on regional water resources assessment models under stationary and changing climate. Water Resour. Manage. 18, 591–612.
- Xu, C.Y., Gong, L.B., Jiang, T., Chen, D.L., Singh, V.P., 2006. Analysis of spatial distribution and temporal trend of reference evapotranspiration and pan evaporation in Changjiang (Yangtze River) catchment. J. Hydrol. 327, 81–93.
- Xu, H., Taylor, R.G., Xu, Y., 2011a. Quantifying uncertainty in the impacts of the climate change on river discharge in sub-catchments of the Yangtze and Yellow River basins, China. Hydrol. Earth Syst. Sci. 15, 333–344.
- Xu, K.H., Milliman, J.D., 2009. Seasonal variations of sediment discharge from the Yangtze River before and after impoundment of the Three Gorges Dam. Geomorphology 104 (3-4), 276–283.
- Xu, K.H., Milliman, J.D., Yang, Z.S., Xu, H., 2007. Climatic and Anthropogenic Impacts on the Water and Sediment Discharge from the Yangtze River (Changjiang), 1950–2005. In: Gupta, A. (Ed.), Large Rivers: Geomorphology and Management. John Wiley & Sons, pp. 609–626.
- Xu, J.J., Yang, D.W., Yi, Y.H., Lei, Z.D., Chen, J., Yang, W.J., 2008. Spatial and temporal variation of runoff in the Yangtze River basin during the past 40 years. Quat. Int. 186 (1), 32–42.
- Xu, Q.X., Yuan, J., Wu, W.J., Xiao, Y., 2011b. Fluvial processes in the middle Yangtze River after impoundment of Three Gorges Project. J. Sediment Res. 4 (2), 38–46 (in Chinese with English abstract).
- Xu, X.B., Tan, Y., Yang, G.S., 2013a. Environmental impact assessments of the Three Gorges Project in China: issues and interventions. Earth-Sci. Rev. 124, 115–125. Xu, Z.G., Zhao, Y.L., Li, B., Zhang, M.W., Wang, Y., 2013b. Forecast of *Microtus fortis*
- disaster in Dongting Lake regions of China. Chin. J. Ecol. 32 (10), 2830–2836. Yang, H.F., Yang, S.L., Xu, K.H., Milliman, J.D., Wang, H.J., Yang, Z.S., Chen, Z.Y., Zhang,
- Yang, H.F., Yang, S.L., Xu, K.H., Milliman, J.D., Wang, H.J., Yang, Z.S., Chen, Z.Y., Zhang, C.Y., 2018. Human impacts on sediment in the Yangtze River: a review and new perspectives. Global Planet. Change 162, 8–17.
- Yang, S.L., Gao, A., Hotz, H.M., Zhu, J., Dai, S.B., Li, M., 2005. Trends in annual discharge from the Yangtze River to the sea (1865–2004). Hydrol. Sci. 50 (5), 825–836.
- Yang, S.L., Milliman, J.D., Li, P., Xu, K., 2011. 50000 dams later: erosion of the Yangtze River and its delta. Global Planet. Change 75, 14–20.

- Yang, S.L., Shi, Z., Zhao, H.Y., Li, P., Dai, S.B., Gao, A., 2004. Effects of human activities on the Yangtze River suspended sediment flux into the estuary in the last century. Hydrol. Earth Syst. Sci. 8 (6), 1210–1216.
- Yang, Z.S., Wang, H.J., Saito, Y., Milliman, J.D., Xu, K.H., Qiao, S., Shi, G., 2006. Dam impacts on the Changjiang (Yangtze) River sediment discharge to the sea: the past 55 years and after the Three Gorges Dam. Water Resour. Res. 42, W04407. https://doi.org/10.1029/2005WR003970.
- Ye, X.C., Li, X.H., Liu, J., Xu, C.-Y., Zhang, Q., 2014. Variations of reference evapotranspiration and its contributing climatic factors in the Poyang Lake catchment, China. Hydrol. Process. 28, 6151–6162.
- Zhang, E.F., Savenije, H.H.G., Chen, S.L., Chen, J.Y., 2012a. Water abstraction along the lower Yangtze River, China, and its impact on water discharge into the estuary. Phys. Chem. Earth. 47–48, 76–85.
- Zhang, J., Wu, Y., Zhang, Y.Y., 2015. Plant nutrients and trace elements from the Changjiang watershed to the East China Sea. In: Zhang, J. (Ed.), Ecological continuum from the Changjiang (Yangtze River) watersheds to the East China Sea continental margin. Springer, Switzerland. https://doi.org/10.1007/978-3-319-16339-0 5.
- Zhang, Q., Liu, Y., Yang, G.S., Zhang, Z.X., 2011. Precipitation and hydrological variations and related associates with large-scale circulation in the Poyang Lake basin, China. Hydrol. Process. 25, 740–750.
- Zhang, Q., Li, L., Wang, Y.G., Werner, A.D., Xin, P., Jiang, T., Barry, D.A., 2012b. Has the Three-Gorges Dam made the Poyang Lake wetlands wetter and drier? Geophys. Res. Lett. 39, L20402. https://doi.org/10.1029/2012GL053431.
- Zhang, Q., Zhou, Y., Singh, V.P., Chen, X.H., 2012c. The influence of dam and lakes on the Yangtze River streamflows: long-range correlation and complexity analyses. Hydrol. Process. 26, 436–444.
- Zhang, Q., Peng, J.T., Xu, C.Y., Singh, V.P., 2014. Spatiotemporal variations of precipitation regimes across the Yangtze River Basin, China. Theor. Appl. Climatol. 115, 703–712.
- Zhao, J., Guo, L.C., He, Q., Wang, Z.B., van Maren, D.S., Wang, X.Y., 2018. An analysis on half century morphological changes in the Changjiang Estuary: spatial variability under natural processes and human intervention. J. Mar. Syst. 181, 25–36.
- Zhao, S.Q., Fang, J.Y., Miao, S.L., Gu, B., Tao, S., Peng, C.H., Tang, Z.Y., 2013. The 7decade degradation of a large freshwater lake in the central Yangtze River, China. Environ. Sci. Technol. 39, 431–436.
- Zhao, Y.F., Zou, X.Q., Liu, Q., Yao, Y.L., Li, Y.L., Wu, X.W., Wang, C.L., Yu, W.W., Wang, T., 2017. Assessing natural and anthropogenic influences on water discharge and sediment load in the Yangtze River, China. Sci. Total Environ. 607–608, 920–932.
- Zheng, S.R., 2016. Reflections on the Three Gorges Project since its operation. Engineering 2, 389–397.
- Zhou, J.J., Zhang, M., Lu, P., 2013. The effect of dams on phosphorus in the middle and lower Yangtze river. Water Resour. Res. 49, 3659–3669.
- Zhou, Y.Q., Jeppesen, E., Li, J.B., Zhang, Y.L., Zhang, X.P., Li, X.C., 2016. Impacts of Three Gorges Reservoir on the sedimentation regimes in the downstreamlinked two largest Chinese freshwater lakes. Sci. Rep. 6, 35396. https://doi.org/ 10.1038/srep35396.
- Zong, Y.Q., Chen, X.Q., 2000. The 1998 flood on the Yangtze, China. Nat. Hazards 22, 165–184.